Author’s response to reviews

Title: A New Methodology for Assessment of the Performance of Heartbeat Classification Systems

Authors:

John M Darrington (jmd@csse.uwa.edu.au)
Livia C Hool (lhool@cyllene.uwa.edu.au)

Version: 2 Date: 15 November 2007

Author’s response to reviews:

* We have amended the paper’s title as suggested.

* The record numbers of each group from the MIT-BIH database have been explicitly stated, as suggested.

* Footnotes have been added to the non-zero entries in Table 2, to make explicit the source of the figures.

* The text "Data 10 and 11" have been replaced with "Datum 10 and Datum 11" as suggested.

* A "Competing Interests" section has been inserted.